Author’s response to reviews

Title: ‘Modified STOPP-START criteria for Sri Lanka’; Translating to a resource limited healthcare setting by Delphi consensus

Authors:

Nithushi R Samaranayake (nithushi@sjp.ac.lk)
A Balasuriya (balasuriya.ac@gmail.com)
GH Fernando (gitaf16@gmail.com)
D Samaraweera (dnsamaraweera@yahoo.com)
LGT Shanika (thushani@sjp.ac.lk)
JKP Wanigasuriya (kamaniw@sjp.ac.lk)
CN Wijekoon (nimalawijekoon@sjp.ac.lk)
CA Wanigatunge (caw@sjp.ac.lk)

Version: 2 Date: 24 Sep 2019

Author’s response to reviews:

Attached separately